Applied Mathematics and Mechanics (English Edition)

https://doi.org/10.1007/s10483-019-2427-9

Thirty years of turbulence study in China*

Bohua SUN[†]

Institute of Mechanics and Technology, Xi'an University of Architecture and Technology, Xi'an 710055, China (Received Sept. 3, 2018 / Revised Oct. 24, 2018)

Abstract This paper provides a comprehensive introduction to the major progress in transition and turbulence studies in China over the past thirty years. Despite the author's best efforts, there are some unavoidable omissions. We expect that with the continued deepening of turbulence research, Chinese scholars will make increasing contributions in this field, which will certainly have strong influence on the establishment of a China School of Turbulence.

Key words turbulence, transition, wall bounded flow

Chinese Library ClassificationO357.52010 Mathematics Subject Classification76F06, 76D10, 76D17, 76F40, 76K05

1 Introduction

Turbulence is everywhere, controlling the drag on cars, airplanes, and trains whilst affecting weather through its influence on large-scale atmospheric and oceanic flows. Even solar flares are a manifestation of turbulence as they are triggered by vigorous motions on the surface of the Sun. It is easy to be intrigued by a subject so pervasive in daily life.

Turbulence is one of the most difficult domains in nonlinear physics owing to its complex, daunting mathematical descriptions, profound difficulties of its inherent instabilities, and its chaotic processes. Until today, there has been no obvious breakthrough in the existence and uniqueness of turbulence solutions in mathematics. However, Chinese scholars have made some promising achievements in turbulence studies over the past 30 years. This article highlights some of the major developments that Chinese scholars have pioneered in the fields of turbulence theory, transition, and calculations, as well as in experiment and measurement techniques.

The paper is organized as follows. Section 2 reviews turbulence theory. Section 3 analyzes turbulence transition. Section 4 examines turbulence calculations. Section 5 evaluates experiment and measurement techniques. Finally, Section 6 concludes with future perspectives.

2 Turbulence theory

2.1 Zhou's turbulence theory

Zhou's (or Chou's) turbulence theory^[1-2] mainly consists of two parts, which can be applied

^{*} Citation: SUN, B. H. Thirty years of turbulence study in China. Applied Mathematics and Mechanics (English Edition), 40(2), 193–214 (2019) https://doi.org/10.1007/s10483-019-2427-9

[†] Corresponding author, E-mail: sunbohua@xauat.edu.cn

Project supported by the National Natural Science Foundation of China (Nos. 10921202, 11221061, 11632002, 11521091, 11602005, and 91752000)

[©]Shanghai University and Springer-Verlag GmbH Germany, part of Springer Nature 2019

to turbulence models with arbitrary high-order statistics and to the statistical theory of turbulent vortex structures by first solving and then averaging. The first part of the theory was based on two studies conducted from 1940 to 1945, while the second one dealt with systematic studies during the 1950s and 1980s. It is worth noting that these two parts led to a number of advances both before and after Zhou's death in 1993, and constituted an important achievement in the field, namely, the basic theory of turbulence in China over the past 30 years.

Zhou's two 1940s studies proposed a step-by-step approximation method to solve the velocity correlation equations at various orders. Using this method and the quasi-similarity condition in the 1980s, one of Zhou's students, Chen, completed the calculations for the channel flow and the plane wake flow, thereby obtaining the result^[3-4] for the third-order correlation. This result was consistent with experimentation. In 1988, another Zhou's student, Meng, completed the calculation for a plane mixing layer and again obtained results that were consistent with experimentation^[5]. In 1991, Lin (also a student of Zhou) determined the relationship between the free shear turbulence and fourth-order correlation in a two-dimensional mixed layer, a turbulent wake, and a turbulent jet^[6]. Comparison of these results with those of previous studies showed that Zhou's theory was successful for these flows. Lin also obtained results for turbulence in a channel using the quasi-similarity theory whilst retaining the viscous term^[7].

Although credible results can be obtained, the odd-order truncated step-by-step approach is complicated and requires certain assumptions to be made at each level, which greatly restricts the theory's extension. Zhou had been thinking about this problem for some time, and in 1988 he proposed a step-by-step iterative method to simultaneously solve the equations for mean flow and turbulent velocity fluctuations^[8].

The following provides a basic understanding of this step-by-step iteration method. First, in the averaged equation of motion and in the second-order velocity correlation equation, the third-order velocity correlation is ignored. The results for the Reynolds stress and the mean velocity are obtained as zeroth-order approximations. These zeroth-order approximation results are then substituted into the turbulent fluctuation equation, which is solved for the fluctuating velocity. The Reynolds stress is then recalculated using this fluctuating velocity. Following this, the new Reynolds stress result is substituted into the averaged equation of motion, which is solved for the mean velocity. Finally, the new Reynolds stress and new mean velocity are taken as the first-order approximations. By repeating this procedure, it is possible to calculate the higher-order approximations of the mean velocity, as well as the velocity dependence. In particular, once the turbulent fluctuating velocity has been obtained, any higher-order velocity correlation can be calculated in principle. It is clear that the calculations in this step-bystep iteration method are simple compared to those in the previous step-by-step approximation method. Another attractive feature of this method is that the fast variables, such as the average velocity of the turbulent flow, the fluctuating kinetic energy, and the fluctuating velocity, are separated because of the assumption of the quasi-similarity theory and fast Fourier transform (FFT), which can be used to solve the fluctuating field in micro-scale normalized regions under a local reference system.

Using this method, Zhou, Huang, and their students conducted numerical calculations of various typical flow fields. Assuming quasi-similarity, Wang et al.^[9] used the FFT to solve the fluctuating field of homogeneous isotropic turbulence, and re-obtained the second- and third-order velocity correlation functions from late decay to a low Reynolds number. For a plane wake, Lin et al.^[10] used the calculated Reynolds stress equivalent as the zeroth-order approximation and applied the FFT to solve the fluctuating velocity field in a local reference system. Thus, they obtained satisfactory results for the fourth-order velocity correlation without any iteration. Furthermore, Lin et al.^[11] used this method to calculate high-order velocity correlations in a channel flow, and again obtained satisfactory results. Meng et al.^[12] obtained satisfactory high-order velocity correlation results for plane mixing layers in the doctoral dissertation. Fan et al.^[13] presented the results of calculations for a plane turbulent jet.

There have been recent developments in the statistical theory of turbulent vortex structures. Tsai and Lin^[14] proposed a new method, which combined Zhou's theory of vortex structures and Kraichnan's direct interaction theory. Considering that Kraichnan's direct interaction theory applies to waves rather than vortices, they first defined a single turbulent vortex as a moving fluid group with a self-similar structure. Following this, they solved the free vortex and used methods derived from quantum field theory to describe the vortex in terms of creation and annihilation operators. The "Schrödinger equation" for vortex interaction was constructed using the interaction term in the Navier-Stokes equation to describe the state change.

In the early 1990s, Huang and his collaborators observed that the spherical vortices of Beltrami flow, obtained as a constant solution of the Euler equation or from decay of the Navier-Stokes equation, can produce complex chaotic phenomena by superposition. In addition, they found an interesting double-ring vortex structure^[15]. They classified the exact vortex solutions described in the literature and found that a straight spiral vortex can be superimposed on an arbitrary coaxial Beltrami flow (again obtained as a constant solution of the Euler equation or from decay of the Navier-Stokes equation), citing Moffatt's well-known spherical vortex as a special case^[16]. Huang^[17] and Huang and Luo^[18] also took a non-axisymmetric vortex as a turbulence element and introduced a general mathematical formula for the direction average, thereby improving the work of Saffman and Pullin^[19]. As the longest-standing collaborator in Zhou's turbulence studies, Huang and Hu^[20] made a systematic summary of the statistical theory and recent developments in the turbulent vortex structure in 2002. This summary was based on Zhou's theory. The closure of the Reynolds equation obtained from the Navier-Stokes equation and Reynolds equation was also briefly discussed by Lee and Lian^[21].

In the 1990s, Zhou's goal of solving the averaged equation of motion and the equation for turbulent fluctuations was finally realized. The statistical theory of the turbulent vortex structure, based on solving following averaging, led to the foundation of a unique approach to turbulence research in China.

For important cases of high-order statistics that arise in engineering, the current popular mode theory cannot go beyond three orders of correlation. In this situation, Zhou's turbulence theory can play an important role. However, how this approach can be used in practice for more challenging turbulent flows (such as wall turbulence and separation flows) remains debatable. By mining direct numerical simulation (DNS) databases as deeply as possible, it is hoped that a combination of Zhou's theory and large eddy simulation (LES) may produce more favorable results.

2.2 Physical mechanisms of homogeneous isotropic turbulence

The study of the physical mechanisms of homogeneous isotropic turbulence is the most important and fundamental aspect of turbulence physics. Revealing and depicting these mechanisms are also the starting point of turbulence modeling.

Intermittency is an important characteristic of turbulent fields (in contrast to random fields) and has always been at the core of turbulence research. Studies that deal with intermittency in turbulence have concentrated on the singular scaling law of turbulence, with the degree to which the scaling index deviates from that of the Kolmogorov K41 theory used as a criterion for intermittency. In the history of theoretical turbulence research, several phenomenological models have been proposed to describe the singular scaling law of turbulence. However, no model to date provides complete information on turbulent structures. There are many reasons for this, one of which is a lack of effective means to identify such structures.

Chen and his collaborators improved the global zero-crossing and peak-valley recognition methods and proposed a local zero-crossing method, which used the number of crossing points as a marker for structural classification. Through analysis of incompressible turbulence, compressible turbulence, and transport of a passive scalar field, they showed that the probability of structural classes, classified by the local zero-crossing method, is scale-invariant in the inertial region. Moreover, the strongest intermittent structure is determined by the structural fluctuations of a single zero-crossing point. Thus, for the first time, structural information was introduced into turbulent batch models. From the analysis of the dynamics of vortex filaments, the above phenomena could be explained. Furthermore, the relations among the law of motion of a vortex filament, the structure of a single zero-crossing point, and the correlations of the strongest intermittent structure were revealed^[22].

In recent years, high-Mach-number turbulence has attracted much attention owing to the needs of the aerospace industry and high-energy-density physics research. The study of multi-scale and cascade processes has always been at the core of turbulence theory. However, there have been few multi-scale analyses from the point of view of multi-process decomposition, which explains the gaps in the understanding of the physical mechanisms of turbulence at a high Mach number. Research by Chen's team clearly showed that the multi-scale analysis must be based on the multi-process decomposition^[23–24]. Thus, a framework for understanding multi-scaling behavior and energy level series analysis in shear/bulging processes was proposed. This proved to be a feasible starting point for the theoretical study of high-Mach number turbulence. This study clearly revealed the physical essence of compressible turbulence at a high Mach number for the first time. It is clear that the modeling of high-Mach-number compressible turbulence should be based on the modeling of both the shear and bulging processes coupled together.

2.3 Spatiotemporal correlation model

Spatiotemporal correlation is the main method for studying the dynamic coupling of the spatial and temporal scales of motion in turbulence^[25-26]. Moreover, it determines the relationship</sup> between a transient turbulence disturbance in one position and an instantaneous disturbance in another position, thereby describing the behavior of turbulent disturbances on both space and time scales. Spatiotemporal correlation models of turbulence include the random scanning model and the local strain model of isotropic and uniform turbulence, the elliptical model of Taylor's freezing and turbulent shear flow, and the linear wave propagation and compressible turbulence models. He and Zhang^[27], He et al.^[28], Zhao and He^[29], and He et al.^[30] proposed an elliptical approximation model of Lagrange space-time correlations that included flow deformation factors. These factors modified Taylor's freezing hypothesis, which essentially represents degeneration of macro-scale vortices in turbulent shear flow [27-30]. The elliptical approximation model was verified both numerically and experimentally, for example, via DNS of turbulent channel flow at a low Reynolds number^[29,31] and experimental results on the Rayleigh-Bénard convection^[30]. The Taylor freezing hypothesis, as the first spatiotemporal correlation model, is widely used in all fields of turbulence research. For example, the spatial energy spectrum can be obtained from continuous-time measurements at a single point in space using this hypothesis. Indeed, many models of turbulent noise are dependent on Taylor's freezing hypothesis. The elliptic model is hugely significant to these related fields.

Spatiotemporal correlations can be used to provide the time scale and precise subgridscale turbulence models for $\text{LES}^{[32-33]}$. In contrast, the simultaneous-time precise subgrid-scale LES method can also be used to study spatiotemporal correlations^[34-35]. The turbulent noise problem^[36] and the particle-loaded turbulence correlation problem^[37] are based on the Eulerian spatiotemporal correlation of Lighthill's acoustic analogy and the Lagrangian time correlation. Hence, the accurate prediction of spatiotemporal correlations is the key to the study of these related problems.

Turbulent coherent structures play an important role in the study of turbulence as their dynamics depend on their specific characteristics. The various methods that identify turbulent structures based on the iso-surfaces of various physical quantities in the instantaneous flow field have clear defects. Thus, Yang et al.^[38] developed and applied a stable and topologically preserved particle inverse tracking method to perform multi-scale geometric analysis of isotropic turbulence. Later, a new definition^[39–40] was proposed wherein the vortex surface was defined by the equivalent surface of the vortex surface field. An associated vortex surface evolution equation was then obtained. The related phenomena of eddy dynamics, including vor-

tex reconnection^[41–42], can thereby be explained. Yang and Pullin^[43–45] and Zhao et al.^[46–47] also studied and compared the Lagrangian and Eulerian structures in the channel flow using computational methods.

2.4 Qian's method

A number of researchers have sought to improve on the Kolmogorov^[48–49] (K41) analysis by matching solutions valid in the inertial subrange to those valid in the viscous dissipation range, i.e., considering finite Reynolds number effects on K41. Among them is Qian, who has published 27 single-authored international peer-reviewed papers dealing with aspects of the scaling properties of homogeneous, isotropic turbulence since 1983. An overview of "Qian's method" (see Refs. [50]–[52]) appears in McComb (2014, pp. 168–170)^[53]. Notably, Qian has investigated whether the second-order exponent corresponds to normal Kolmogorov scaling or anomalous scaling. Qian's method makes use of exact relationships incorporated with wellestablished data correlations available in the literature to extract as much physics as possible from the available experimental results.

2.5 Quantifying wall turbulence via a symmetry approach — a Lie-group theory

She et al.^[54], Chen et al.^[55], and Chen and She^[56] recently revealed the first unified analytic description of the non-uniform distribution of mean velocity and turbulence intensities across the entire domain of turbulent boundary layers for all Reynolds numbers. This had been a century-old goal dating back to Prandtl's pioneering work in 1904. This Lie group inspired the symmetry approach. Their method is particularly interesting in terms of methodology, as they proposed a universal dilation-symmetry-breaking form, which, as they argued, is necessarily present in non-equilibrium and complex systems (such as turbulence). In other words, they extended the classical symmetry analysis, which usually assumes symmetry preservation to deal with a non-equilibrium system with symmetry-breaking. Indeed, this method was successfully applied to describe a difficult problem involving rough pipe^[57], for which the roughness near the wall added a new dimension to the complexity and compressible flow with density and temperature variations^[58–59].

3 Turbulence transition

Luo et al.^[60] studied the effects of disturbances on both the basic flow profile and the transition process in the plane channel flow. They found that the growth of a disturbance can correct the basic flow, leading to a rapid increase in the range of wavenumbers of the unstable disturbance and a rapid increase in the growth rate. This causes the "breakdown" process. The results of this study revealed the importance of corrections to the basic flow and of the characteristics of nonlinear interaction between disturbances to the transition process. Zhou's research group then studied the systematic mechanisms. Tang et al.^[61] presented results for an incompressible boundary layer. Huang et al.^[62] performed a time model calculation and simulation for the mechanism of the laminar flow catastrophe in a supersonic boundary layer transition. Similar results were presented by Cao et al.^[63]. For a supersonic plate boundary layer, Dong and Zhou^[64] observed the same mechanism in the bypass transition, which was caused by large fluctuations of the incoming flow. These results confirmed that basic flow correction can promote the transition process. Zhang and Zhou^[65] studied the boundary layer receptivity of vortex perturbations in the free stream, while Gao et al.^[66] studied the receptivity of fast and slow sound waves in a supersonic boundary layer. The latter showed that a disturbance in the free stream can stimulate an unstable disturbance in the boundary layer through the outer edge. This provided a new way to understand the receptivity of the free stream disturbance in the boundary layer. Ren and Fu^[67] studied unsteady disturbance spectrum in a boundary layer with a curved surface at Mach 4.5. They found that a flow vortex/entropy disturbance could be transformed into a quasi-steady Görtler vortex, while a slow sound wave could be converted into an unsteady Görtler vortex. Jiang and Lee^[68] reviewed the receptivity of hypersonic boundary layers. In addition, Ren and Fu^[69] studied secondary instability of Görtler vortices and found the anti-symmetric mode to be more unstable than the symmetric mode at medium and high Mach numbers. Ren et al.^[70] used the parabolic stability equation (PSE) method to study the effects of streaks and Görtler vortices on the first and second modes, and found that the proper amplitude of the streak and the Görtler vortex can inhibit the growth of the first and second modes, thereby postponing the transition.

The e^N method is considered to be a transition prediction method with a relatively firm theoretical basis and has mostly been used in the aviation industry. However, this method has two weaknesses. The receptivity problem is not considered, and the geometry of the transition position is determined entirely by experience. A study by Su and Zhou^[71–72] showed that the traditional e^N method has problems in predicting transitions. Thus, they considered receptivity and proposed an improved e^N method. The results of this method are consistent with both numerical simulation results and those of wind tunnel experiments^[73–74].

Li et al.^[75] studied the natural transition in a blunt cone boundary layer at Mach 6 and at a small angle of attack using DNS. They found that low-frequency disturbances are more significant in the late stages of transition than in the second mode (see Fig. 1). Liang et al.^[76] used DNS to study the effect of wall temperature on stability in a blunt cone boundary layer at Mach 8. They found that reducing the wall temperature can significantly increase secondmode instability but can inhibit first-mode instability. In terms of the relationship between the characteristics of the flow disturbance and the transition position, Su and Zhou^[77] studied the changes in the transition position caused by changes in the disturbance amplitude over time. They found that the change in the transition position lags when the amplitude of the upstream disturbance changes over time. Both the turbulent and laminar flows appear alternately.

Liu and Luo^[78] studied the effects of different frequency disturbances on the transition position in a hypersonic blunt cone boundary layer at a small angle of attack. They found that the transition position depends on the distribution of the perturbation amplitude with frequency, which could explain the shift or forward movement of the transition position in a hypersonic blunt cone boundary layer.

These studies illustrate how the transition process depends strongly on the characteristics



Fig. 1 Distribution of the skin fraction coefficient on the surface of a cone from Li et al.^[75], (a) three-dimensional space and (b) $z\theta$ -plane (color online)

of both flow disturbances and wall disturbance, and reveals the scope for a wide-ranging investigation of the entire transition process.

The PSE approach is an efficient method used to calculate the evolution of perturbations in a boundary layer. Li and Luo^[79] studied these transitions in an incompressible boundary layer using this method. Zhang and Zhou^[80–81] applied the PSE method to study transitions in a compressible and supersonic boundary layer. All these investigations produced favorable results. In addition, rapid calculation methods combining the PSE and DNS were proposed to study both transitional and fully developed turbulence, which allowed for effective prediction of the evolution of perturbations and the transition to turbulence in supersonic and hypersonic boundary layers. Zhang and $Su^{[82]}$ proposed an extension of the PSE method, and compared their results with those of the traditional PSE. They found little difference between them. Yu et al.^[83] proposed a stability analysis method called the EPSE. This method is also based on an eigenvalue problem; however, it can handle the stability analysis of nonparallel basic flows, thereby widening its range of application. Based on the PSE theory, Luo et al.^[84] developed a highly efficient numerical method to solve the nonlinear evolution of finite-amplitude Tollmien-Schlichting (T-S) waves, and analyzed the new modes produced by the nonlinear interaction of disturbance waves. Using the linear stability theory and multiple scales method, Tang et al.^[85] showed the influence of nonparallel flow on boundary layer stability. The stability problem of three-dimensional T-S waves in a non-parallel boundary layer was analyzed using the PSE method with the original-variable form of three-dimensional disturbance waves^[86]. Guo et al.^[87] analyzed and compared the multiple unstable modes of inviscid and viscous flows at a high Mach number through a numerical approach. Finally, Liu et al.^[88] studied the evolution of C-type instability in a nonparallel boundary layer using the PSE method. A systematic summary of the transition process, transition mechanism, and transition control in boundary layers was given in Ref. [89].

The research team headed by Lee at Peking University has also completed much experimental and theoretical work on hypersonic boundary layer transitions^[68,90]. Zhang et al.^[91] and Zhang and Lee^[92] were the first in China to publish quiet wind tunnel experimental results. In particular, they used a CO₂ Rayleigh scattering technique to obtain clear images of the hypersonic laminar-turbulent transition, as shown in Fig. 2. Tang et al.^[85] studied the effect of two-dimensional roughness elements on the evolution of second-mode waves by means of the partially coherent beam (PCB) technique, particle image velocimetry (PIV), and other experimental methods. Zhang et al.^[93] and Zhu et al.^[94] successfully applied PIV to hypersonic boundary layer measurement and found that the second-mode waves first increased and then decayed, whilst the low-frequency disturbance amplitudes were large. Zhu et al.^[95–96] performed experiments and numerical simulations at Mach 6 in a cone boundary layer and studied aerodynamic heating, which was highly praised by Sun and Oran^[97]. They found that the bulging effect of second modes during the transition process can cause large rises in temperature, which can exceed those of turbulent-laminar transition effects. Chen et al.^[98] used the PSE, the two-time instability theory, and an energy analysis method to study the interaction



Fig. 2 CO_2 flow visualization of the boundary layer transition on a flared cone^[91]

between second and first modes in a conical boundary layer at Mach 6. They found that the second mode played the role of a catalyst, causing a rapid increase in the first mode.

4 Turbulence calculations

In recent years, the computational speed of computers has rapidly increased, and the development of supercomputers and parallel computing methods have provided better conditions for numerical simulations of turbulence. DNS provides direct solutions for the Navier-Stokes equation of control flow, meaning that the information on all turbulence scales can be obtained. DNS is one of the key methods currently used for the study of turbulence. The Navier-Stokes equations for both incompressible and compressible fluids were studied by Li et al.^[99–100] using DNS. Li et al.^[101] used DNS to study coherent structures in a Mach 6 flat-plate boundary laver. They found that quasi-directional vortices dominated at high Mach numbers, whereas entrapment vortices rarely appeared^[102]. In particular, OpenCFD, an open source compressible high-precision computational fluid dynamics software that Li developed, has been widely recognized and applied. Deng and Zhang^[103] developed a weighted compact nonlinear scheme (WCNS). Ma and Fu^[104] developed a group velocity control (GVC) scheme. Both schemes showed good performance. Dong et al.^[105], Dong and Luo^[106], Dong and Zhou^[107], Dong and Luo^[108], and Dong^[109] conducted a series of turbulent boundary layer studies under supersonic and hypersonic conditions using DNS and the Baldwin Lomax model. Finally, Dong and Zhou^[110] accomplished corrections to the eddy thermal conductivity near the peak of turbulent kinetic energy.

In addition, the gas kinetic method is a new approach based on the kinetic theory of gases. Li and Fu^[111], Li et al.^[112], Tan et al.^[113], and Fu and Li^[114] applied this method to DNS of compressible turbulence. Their results showed that the high-order kinetic scheme is superior to the second-order scheme with regard to capturing the fine structure of flow fields (such as in near-wall turbulence). However, the computational complexity is significant. Further research should be conducted to optimize the reconstruction quality and to improve the efficiency of this scheme.

Turbulence models are a key element in CFD and have a decisive influence on the accuracy of numerical simulations. A good turbulence model allows for significant savings on experimental costs. At a low Reynolds number, a nonlinear eddy viscosity model was established on the basis of the FRT model, which itself is based on the models of Fu and $\text{Guo}^{[115]}$. This Fu-Rung-Thiele (FRT) model does not contain wall parameters and is suitable for turbulence simulations and complex flow fields. In compressible turbulence, the fluid exhibits strong anisotropy. Based on a theoretical analysis of the dynamic pressure and fluctuating velocity fields in compressible turbulence and on DNS results, Zhang and Fu^[116] formulated a compressibility correction model for correlation of the pressure deformation rate in compressible and uniform shear turbulence.

LES has progressed significantly in recent years, and is likely to become the principal method for turbulence simulation in the future. Peking University has completed some pioneering work on the LES of turbulence. First, a constrained LES was proposed. Physical systems are subject to a number of fundamental conservation laws and symmetries. In turbulence, these are manifested as a set of physical constraints. The idea of constraints in this context can be traced back to Kraichnan and Chen's 1987 decimation theory^[117]. In addition, Meneveau^[118] enumerated the statistical relationships to be satisfied by the LES subgrid stress model. In the application of LES, these physical constraints are often destroyed. By introducing physical constraints, the Peking University team was able to solve a series of previously intractable problems in LES.

The dynamic subgrid scale stress model is currently considered to be a promising LES model. It takes advantage of the mathematical properties of the subgrid stress in the turbulent field. The model coefficient is determined by a variational principle, with no need for human

intervention. After the kinetic model was proposed in the 1990s, it received great attention and has been widely used.

Chen's team extended the variational principle of the dynamical model to a constrained variational principle by introducing constraints, thereby solving the problem of imprecise simulation of the energy spectrum in a hybrid injection dynamics model^[119]. This work is considered to be an important development of the kinetic model^[120].

At present, the main challenge that LES faces is the simulation of near-wall turbulence at a high Reynolds number. To avoid high cost incurred in calculating near-wall turbulence in an LES simulation, the Reynolds-averaged Navier-Stokes (RANS)/LES method was proposed. This method combines the advantages of the RANS method with those of LES, while avoiding their respective defects. A hybrid RANS/LES method, called the detached eddy simulation (DES) method, has been successfully applied to aerodynamic simulation and is considered to be one of the most promising methods for turbulence simulation.

At present, the RANS/LES method is in its infancy, the most important remaining problem being the lack of small-scale fluctuations in the near-wall simulation, leading to the emergence of a nonphysical transition zone between the RANS simulation area and the LES simulation area and deleterious effects on simulation accuracy.

Chen et al.^[121] proposed a Reynolds stress constraint on the near-wall LES model. This method produces sufficient small-scale fluctuations in the near wall simulation, thereby dealing with the shortcomings of the traditional RANS/LES mixing method. They also proposed an advanced turbulence model that can be applied to aerodynamic simulations.

Constrained LES (CLES) has been applied to the simulation of a number of aerodynamic problems. The CFD team at Peking University used independent BXCFD software in 2012 to implement CLES for simulation of the flow around a large airliner and revealed a more complex flow structure than the DES. Not only does CLES provide higher precision in aerodynamic simulations than those achievable with DES, it also provides a fine vortex structure, which can be used as a source of aerodynamic noise to provide a more accurate simulation of the aircraft noise during take-off and landing. This study was a cover story in *Science China*^[122].

As previously mentioned, the CLES method solves the nonphysical interface problem that plagues RANS/LES, and, in addition, the simulation accuracy is clearly better than that of the DES method, which was developed by Boeing. At the 23rd World Congress of Mechanics that was held in August 2012, Chen et al.^[123] made an opening presentation entitled multi-scale fluid mechanics and modeling. The fluid mechanics and turbulence portions of this report describe the introduction of an energy dissipation constraint into the dynamic model and a Reynolds stress constraint into the high-Reynolds-number wall turbulence simulation along with the simulation results of the CLES method for a large aircraft. This work received significant attention.

In nature, the atmospheric boundary layer is mostly in a turbulent state. Li and Xie^[124] and Li^[125] studied canopy turbulent flow and developed a transient structure function (TSD) subgrid model. Cui et al.^[126] compared the effects of different numerical methods on predicting urban micro-environmental flow using the LES method. Their results showed that the Lagrangian dynamic model (LDM) is a cost-effective subgrid model for the simulation of complex turbulence. Thus, Cui et al.^[127] proposed a new eddy viscosity model for LES of anisotropic turbulence, which was applied to rotating turbulence and wall turbulence.

Many fluid flows of practical importance have unsteady characteristics on multiple scales. Among studies of complex flows, those around a cylinder are of particular importance. Hence, both Lu's group^[128–130] and Zhou's group^[131–140] investigated unsteady flow in the boundary layer and wake of a cylinder. Li's research group used PIV technology to study the twodimensional flapping flag in a water tunnel, and obtained experimental phenomena such as the period and bifurcation. In addition, they obtained the flow field data around the flag surface^[141]. Zhou's group studied the flows around multiple cylinders and square columns, developing a complex flow theory^[142-144] of practical engineering significance. Xia et al.^[145] studied the dissipation function in turbulent plane Poiseuille flows and plane Couette flows subject to spanwise rotations. Research on multi-scale complex flows also includes the fluid dynamics of swimming and flying organisms^[146], unsteady vortex separation flows, vortex control, hemodynamics and physiological microcirculation, micro scale industrial flows, and heat transfer. Based on CFD analysis, Lu's team studied the propulsion performance of flapping wings and the law of vortex shedding related to biological movement^[147], and improved the understanding of the mechanism of fish swimming in terms of drag reduction and propulsion efficiency^[148].

5 Turbulence experiment and measurement techniques

Since the original Reynolds' experiment, significant contributions have been made to the fluid mechanics community. Because wall turbulence is easier to operate experimentally, it has become a standard geometric model for modern turbulence research. In recent decades, Lee and other researchers have conducted in-depth studies of incompressible boundary layer transitions^[149–153]. Furthermore, Lee improved upon the Hama^[154] experiment and observed the flow structure under a finer natural transition through the parallel setting of hydrogen bubble lines at different heights. For the first time, the soliton-like coherent structure (SCS) (formed at the initial stage of transition) was proposed^[155]. Kachanov observed the periodic "spike (peak)" structure from the quantitative test of hot wire, and confirmed that the "spike" refers to the velocity traces of the chain of ring vortices. Lee obtained the results of the first flow visualization of the chain of ring vortices, which was the result of interaction between the secondary closed vortex and the Λ -vortex (see Fig. 3). This interaction also produced the SCS. These experiments indicated that the SCS is ever-present and dominates the transition process^[149,156–157].



Fig. 3 Formations of (a) the SCS and (b) the Λ -vortex, and (c) different development stages of the SCS and the Λ -vortex

In addition, Lee proposed a new idea to aid understanding of the turbulent transition model and initiated a self-consistent dynamic model for wall turbulence generation. He devised a general framework for the mechanism of the onset of turbulence. The formation, development, and interaction of various coherent structures were explained in detail. Zhao et al.^[158] proved that a solitary-like wave is a common structure in the transitional boundary layer. In their study of wall turbulent flow structure, Lian^[159–160] proposed and explained the important characteristics of the flow structure of the long strip (long streak) and black spot (black spot) in wall turbulence. Guo et al.^[161] used the hydrogen bubble method to visualize the laminar flow turbulent transition in the plate boundary layer flow in the water tunnel. Thus, they discovered the inverse hairpin vortex. Finally, a new method of delayed transition^[162], called the spanwise discrete suction, was developed to produce a high amplitude stable streak.

Gong et al.^[163] conducted a series of work on the relationships between the turbulent energy series, kinetic process, and turbulence occurrence. Using the main dynamic process established in the boundary layer transition, the concept of turbulence cascading was further analyzed. Following this, a quantitative analysis of the measurement results of boundary layer transitions was conducted. From the energy spectrum, the relationship between the turbulent cascade and the dynamic process of the transition was clearly visible^[164]. Moreover, when the energy spectrum is used to analyze the energy level series, it must be clear that the frequency distribution caused by physical or non-physical factors is transmitted to either a low or high frequency. Otherwise, a true and false situation will occur and an error analysis result will emerge^[165]. Through detailed chaotic dynamic analysis of the transition boundary layer, the existence of the attractor in the transition process was confirmed and the relationship between chaos and transition in the boundary layer was established^[166–167].

The intensity of the vortex structure in a turbulent boundary layer is significant for the generation and development of hairpin vortices. Gao et al.^[168] found that there is a significant difference between the real eigenvalue direction and the vorticity vector direction of the identified vortex set in the turbulent boundary layer. Based on the PIV and DNS data, the intensity of three-dimensional swirling flow, which is based on a local velocity gradient tensor, was used to identify the vortex core location^[169–170].

Based on the PIV and hydrogen bubble flow display, Wang et al.^[171], Zhang et al.^[172], and He et al.^[173] used PIV and hydrogen bubble flow visualization to study the effects of roughness on the bypass transition induced by cylindrical wake, the evolution of vortex structure in the boundary layer, and the initial development of the boundary layer disturbance under the effects of the wake. Thus, a new transition control strategy was proposed based on these investigations. In free-stream-induced transition^[174], the two-dimensional cylindrical array-induced transition mode shows differences from the Klebanoff mode. Furthermore, detailed studies show that the evolution dynamics and maintenance mechanism of hairpins and hairpackets are consistent with those in turbulent boundary layers. Wake vortices play an important role in generating and destroying the secondary transverse vortices. Thereafter, the internal mechanism becomes dominant and leads to the establishment of the self-sustaining turbulent boundary layer. The aforementioned study found that the flow of the hairpin vortex determines the logarithm of the turbulent boundary layer^[175–177]. In addition, it was found that the downstream convection of the hairpin vortex determines the transport characteristics of the turbulent boundary layer inside and outside the logarithmic region. The proper orthogonal decomposition (POD) of PIV data indicates a low velocity streak associated with the hairpin pocket and the hairpin vortex in the transitional turbulent boundary layer, which is the main coherent structure near the wall^[178]. Finally, the evolution of the single low-speed streak in the laminar boundary layer^[179] and the gap between understanding the streak characteristics and the dynamic mechanism of the free flow induced bypass transition were also studied.

The reverse transition phenomenon in a flat plate boundary layer was studied by Zhang et al.^[180], while the changes in the turbulent energy generation term, dissipation term, and

absorption term before and after heating were measured. In addition, the physical mechanism of the reversal transition was preliminarily explained.

In addition to wall turbulence, complex flow turbulence has also been the focus of significant research. Important results were obtained concerning circular turbulent jets based on flow visualization and PIV. It was discovered that the development of the flow vortex plays an important role in the wrapping and rolling process of the near field of round turbulent jets. Furthermore, through the study of dynamic excitation control (excitation frequency range, amplitude, and mode) and static control, the deformation degree of the turbulent quasi order vortices is considered to be controllable to a certain degree^[181]. In the study of particle turbulence, owing to the limitation of experimental technology, complete field research was mainly focused on simple cases. At present, the PIV experimental technique for simultaneous measurement of a two-phase flow is a possible breakthrough direction^[182]. Turbulent thermal convection is also a large domain of focus and mainly includes turbulent heat transfer, quasi sequence structure (plume and large scale circulation) in turbulent heat convection, statistics of the turbulent pulse momentum of an RB system, and the study of some non-traditional means (rotation, modulation, stratification, and multi-phase Rayleigh-B nard convection)^[183]. In free surface geomantic boundary experiments, Wang et al.^[184] made a series of important advances. Moreover, Li's group made a breakthrough in the complex flow problem, which was caused by a falling disk. Furthermore, they fully explained the complex vortex structure during the falling process, both theoretically and experimentally [185-186].

Due to increasing aircraft speeds, the transition of the hypersonic compressible boundary layer has become a bottleneck in the aerospace field.

For the study of hypersonic boundary layer transitions, Peking University established two hypersonic quiet wind tunnels. Their team developed temperature sensitive painting (TSP), near wall PIV, Rayleigh scattering imaging (RSI), and so on. The first and second modes of boundary layer transition and the correlation between temperatures were studied. The first flow visualization of the complete transition process of the hypersonic boundary layer was obtained (see Fig. 4), and the two mode wave structures in the hypersonic flow field were clearly captured by the PIV technique for the first time. Furthermore, it was found that the second mode wave generated violent aerodynamic heating at the early stage of transition and that this value may exceed that of the aerodynamic heating once in the developed turbulent state. This process is closely related to the high frequency compression-expansion process, which is caused by the second mode wave and constructs a new principle of aerodynamic heating^[90]. In the experimental domain, Jia et al.^[187–188] produced a successful near wall measurement for a moving wall by means of the PIV method in the air for the first time. Li and Zhang^[189] and Zhang and Li^[190] studied the effect of glow discharge as an artificial disturbance on the first mode and the second mode. Experiment and linear stability results showed that the second-mode instability waves are significantly stimulated by the artificial disturbances. and the boundary layer transition is effectively triggered. Finally, Zhu et al.^[191] developed a new PIV data processing method, which can be used to obtain high accuracy data in a near wall region, and found a strong dilatation process within the second-mode instability by this method^[192]. Testing techniques of nano-tracer planar laser scattering techniques were used to study compressible turbulence^[193]. Clear flow structures were obtained in typical flow fields such as wake flow, supersonic boundary layer transition, and turbulent shock wave boundary layer interaction.



Fig. 4 Complete process of the hypersonic boundary layer transition

Land desertification and dust storms have posed great threats to human environments. Thus, Zheng's team conducted a series of studies on wall turbulence with both air and sand at high Reynolds numbers. During wind-blown sand movement, sand grains are charged to form a wind-blown sand electric field. Moreover, Huang and Zheng^[194] and Zheng et al.^[195] studied the electrification of sand particles in the sand movement, both theoretically and experimentally. Their results showed that the charge of sand particles is mainly produced by collision and friction between grains of different sizes, and the electric field in the wind sand flow is mainly formed by moving charged sand particles. Zheng et al.^[196] proposed a new theoretical model based on the theory of electron/hole exchange to explain the mechanism of the effects of relative humidity on the contact charges of identical insulator particles. Some numerical simulation results showed that the greater the friction wind speed, the smaller the particle size, the greater the calculated concentration at the same height, and the greater the evolution time^[197]. Hence, wind speed and particle size have significant influence on desert formation^[198].

The definition of whirlpools in the flow field is not clear. However, the generation and evolution of vortices in whirlpools play an important role in the flow. It is possible to use a flow control method to clarify the dynamic properties of the whirlpool and to improve the properties of the flow field by inputting small amounts of energy. Hence, Chen et al.^[199] studied the response of disturbance to asymmetric vortices. This work has potential practical significance for the flow control of vortices.

6 Prospects

In the early twentieth century, China's turbulence research was outstanding in the world because of the great work of Peiyuan Zhou. Of course, the great contributions of C. C. Lin and other overseas Chinese were not covered in this article. However, the contributions of Chinese scientists from across the globe, particularly during the second half of the twentieth century, have been included.

In summary, the study of turbulence in China began with Peiyuan Zhou. After more than 70 years, several generations of effort have resulted in some remarkable achievements in the field. With the development of China's economy and the need for national strategic scientific projects, turbulence research in China has been greatly promoted. In 2017, a China national strategic turbulence research program, i.e., Physical Mechanisms of the Formation, Evolution, and Interaction of Turbulence Structures supported by the National Natural Science Foundation of China (No. 91752000), was proposed and initiated under the strong leadership of Professor Shiyi Chen. It can be expected that, with the continued deepening of turbulence research, Chinese scholars will make more and more significant contributions in the field. These contributions will certainly have a strong impact on the establishment of a China School of Turbulence.

Acknowledgements The author would like to acknowledge the support of Cunbiao LEE for his useful help.

References

- CHOU, P. Y. On an extension of Reynolds' method of finding apparent stress and the nature of turbulence. *Chinese Journal of Physics*, 4, 1–33 (1940)
- [2] CHOU, P. Y. On velocity correlations and the solutions of the equations of turbulent fluctuation. *Quarterly of Applied Mathematics*, 3, 38–54 (1945)
- [3] CHEN, S. Y. and ZHOU, P. Y. The application of quasi-similarity conditions in turbulence modeling theory. Advances in Hydrodynamics, 2(2), 1–14 (1987)
- [4] ZHOU, P. Y. and CHEN, S. Y. On the theory of turbulence for incompressible fluids. Sadhana, 10(3/4), 403–417 (1987)

- [5] ZHOU, P. Y., HUANG, Y. N., and MENG, Q. G. The pseudo-similarity theory of the plane turbulent mixing layer. Acta Mechanica Sinica, 22(1), 1–8 (1990)
- [6] ZHOU, P. Y., LIN, J. Z., and WEI, Z. L. The experimental results of 2-d free shear turbulent flows and the comparison with the theoretical computation. *Journal of Hydrodynamics, Ser. B*, 3, 1–9 (1991)
- [7] LIN, J. Z. and HUANG, Y. N. The pseudo-similarity theory of turbulence and its application to a channel flow. *Journal of Hydraulic Engineering*, 12, 72–77 (1994)
- [8] LI, Y. G., QIN, R. X., and ZHOU, P. Y. Laser double-frequency lock-in displacement sensor with application to gravitational radiation detection. *Experimental Gravitational Physics*, World Scientific, Singapore (1988)
- [9] WANG, X. H., HUANG, Y. N., and ZHOU, P. Y. Statistical theory of homogeneous isotropic turbulence for incompressible fluids. *Science in China Series A-Mathematics, Physics, Astronomy* and Technological Science, 37(2), 209–220 (1994)
- [10] LIN, J. Z., HUANG, Y. N., and ZHOU, P. Y. The new method of solving velocity correlation functions and application to the plane turbulent wake. Acta Mechanica Sinica, 9(2), 102–109 (1993)
- [11] LIN, J. Z., HUANG, Y. N., and ZHOU, P. Y. An approximate method for solving turbulent velocity correlation functions in channel flows. *Chinese Science Bulletin*, 40(14), 1167–1172 (1995)
- [12] MENG, Q. G., HUANG, Y. N., and ZHOU, P. Y. The statistical theory of the plane turbulent mixing layer flow (in Chinese). *Journal of Hydrodynamics*, 10(4), 443–450 (1995)
- [13] FAN, M., HUANG, Y. N., and ZHOU, P. Y. The solution for incompressible plane turbulent jet by using the method of successive substitution (in Chinese). *Journal of Hydrodynamics*, 11(4), 465–470 (1996)
- [14] TSAI, S. T. and LIN, D. M. A note on a new theory of turbulence. Applied Mathematics and Mechanics (English Edition), 12(1), 101–105 (1991) https://doi.org/10.1007/BF02018074
- [15] SHI, C. C., HUANG, Y. N., ZHU, Z. X., SU, W. D., and DONG, Y. F. Chaotic phenomena produced by the spherical vortices in the Beltrami flows. *Chinese Physics Letters*, 9(10), 515–518 (1992)
- [16] HUANG, Y. N. and HU, X. Superposition about the 3D vortex solutions of the fluid dynamic equation. Applied Mathematics and Mechanics (English Edition), 21(12), 1359–1370 (2000) https://doi.org/10.1007/BF02459214
- [17] HUANG, Y. N. New development of Prof. Zhou's statistical theory of turbulence (in Chinese). Acta Scientiarum Naturalium, Universitatis Pekinensis, 34(2/3), 151–158 (1998)
- [18] HUANG, Y. N. and LUO, X. P. On orientation averages for vortex models of turbulence. *Physics of Fluids*, **11**(8), 2381–2386 (1999)
- [19] SAFFMAN, P. G. and PULLIN, D. I. Calculation of velocity structure functions for vortex models of isotropic turbulence. *Physics of Fluids*, 8(11), 3072–3084 (1996)
- [20] HUANG, Y. N. and HU, X. The tensor denotation of Beltrami spherical vortices and their symmetry analysis. Applied Mathematics and Mechanics (English Edition), 23(1), 13–17 (2002) https://doi.org/10.1007/BF02437725
- [21] LEE, C. B. and LIAN, Q. X. The closed characteristics of the Reynolds equation. Acta Aeronautica et Astronautica Sinica, 14(5), 335–336 (1993)
- [22] YANG, K., XIA, Z. H., SHI, Y. P., and CHEN, S. Y. Effect of oscillation structures on inertialrange intermittence and topology in turbulent field. *Communications in Computational Physics*, 19(1), 251–272 (2016)
- [23] WANG, J. C., SHI, Y. P., WANG, L. P., XIAO, Z. L., HE, X. T., and CHEN, S. Y. Scaling and statistics in three-dimensional compressible turbulence. *Physical Review Letters*, **108**(21), 214505 (2012)
- [24] WANG, J. C., YANG, Y. T., SHI, Y. P., XIAO, Z. L., HE, X. T., and CHEN, S. Y. Cascade of kinetic energy in three-dimensional compressible turbulence. *Physical Review Letters*, **110**(21), 214505 (2013)

- [25] WU, T., GENG, C. H., YAO, Y. C., XU, C. X., and HE, G. W. Characteristics of space-time energy spectra in turbulent channel flows. *Physical Review Fluids*, 2(8), 084609 (2017)
- [26] HE, G. W., JIN, G. D., and YANG, Y. Space-time correlations and dynamic coupling in turbulent flows. Annual Review of Fluid Mechanics, 49, 51–70 (2017)
- [27] HE, G. W. and ZHANG, J. B. Elliptic model for space-time correlations in turbulent shear flows. *Physical Review E*, 73(5), 055303 (2006)
- [28] HE, G. W., JIN, G. D., and ZHAO, X. Scale-similarity model for Lagrangian velocity correlations in isotropic and stationary turbulence. *Physical Review E*, 80(6), 066313 (2009)
- [29] ZHAO, X. and HE, G. W. Space-time correlations of fluctuating velocities in turbulent shear flows. *Physical Review E*, **79**(4), 046316 (2009)
- [30] HE, X. Z., HE, G. W., and TONG, P. Small-scale turbulent fluctuations beyond Taylor's frozenflow hypothesis. *Physical Review E*, 81(6), 065303 (2010)
- [31] GENG, C. H., HE, G. W., WANG, Y. S., XU, C. X., LOZANO-DURAN, A., and WALLACE, J. M. Taylor's hypothesis in turbulent channel flow considered using a transport equation analysis. *Physics of Fluids*, 27(2), 025111 (2015)
- [32] HE, G. W., RUBINSTEIN, R., and WANG, L. P. Effects of subgrid-scale modeling on time correlations in large eddy simulation. *Physics of Fluids*, 14(7), 2186–2193 (2002)
- [33] YANG, Y., HE, G. W., and WANG, L. P. Effects of subgrid-scale modeling on Lagrangian statistics in large-eddy simulation. *Journal of Turbulence*, 9, N8 (2008)
- [34] HE, G. W., WANG, M., and LELE, S. K. On the computation of space-time correlations by large-eddy simulation. *Physics of Fluids*, 16(11), 3859–3867 (2004)
- [35] GUO, L., LI, D., ZHANG, X., and HE, G. W. LES prediction of space-time correlations in turbulent shear flows. Acta Mechanica Sinica, 28(4), 993–998 (2012)
- [36] YAO, H. D. and HE, G. W. A kinematic subgrid scale model for large-eddy simulation of turbulence-generated sound. *Journal of Turbulence*, 10, N19 (2009)
- [37] JIN, G. D., HE, G. W., and WANG, L. P. Large-eddy simulation of turbulent collision of heavy particles in isotropic turbulence. *Physics of Fluids*, **22**(5), 055106 (2010)
- [38] YANG, Y., PULLIN, D. I., and BERMEJO-MORENO, I. Multi-scale geometric analysis of Lagrangian structures in isotropic turbulence. *Journal of Fluid Mechanics*, 654, 233–270 (2010)
- [39] YANG, Y. and PULLIN, D. I. On Lagrangian and vortex-surface fields for flows with Taylor Green and Kida-Pelz initial conditions. *Journal of Fluid Mechanics*, 661, 446–481 (2010)
- [40] YANG, Y. and PULLIN, D. I. Evolution of vortex-surface fields in viscous Taylor-Green and Kida-Pelz flows. Journal of Fluid Mechanics, 685, 146–164 (2011)
- [41] ZHAO, Y. M., YANG, Y., and CHEN, S. Y. Vortex reconnection in the late transition in channel flow. Journal of Fluid Mechanics, 802, R4 (2016)
- [42] ZHAO, Y. M., YANG, Y., and CHEN, S. Y. Evolution of material surfaces in the temporal transition in channel flow. *Journal of Fluid Mechanics*, **793**, 840–876 (2016)
- [43] YANG, Y. and PULLIN, D. I. Geometric study of Lagrangian and Eulerian structures in turbulent channel flow. Journal of Fluid Mechanics, 674, 67–92 (2011)
- [44] YANG, Y. and PULLIN, D. I. On Lagrangian and vortex-surface fields for flows with Taylor-Green and Kida-Pelz initial conditions. *Journal of Fluid Mechanics*, 661, 446–481 (2010)
- [45] YANG, Y. and PULLIN, D. I. Evolution of vortex-surface fields in viscous Taylor-Green and Kida-Pelz flows. Journal of Fluid Mechanics, 685, 146–164 (2011)
- [46] ZHAO, Y., YANG, Y., and CHEN, S. Evolution of material surfaces in the temporal transition in channel flow. *Journal of Fluid Mechanics*, **793**, 840–876 (2016)
- [47] ZHAO, Y., XIONG, S., YANG, Y., and CHEN, S. Sinuous distortion of vortex surfaces in the lateral growth of turbulent spots. *Physical Review Fluids*, 3(7), 074701 (2018)
- [48] KOLMOGOROV, A. N. The local structure of turbulence in incompressible viscous fluid for very large Reynolds numbers. *Mathematics and Its Applications (Soviet Series)* (ed. TIKHOMIROV, V. M.), Vol. 25, Springer, Dordrecht (1991)

[49]	KOLMOGOROV, A. N. Dissipation of energy in the locally isotropic turbulence. <i>Proceedings</i> :
[50]	QIAN, J. Scaling exponents of the second-order structure function of turbulence. Journal of
[21]	Physics A: Mathematical and General, $31(14)$, $3193-3204$ (1998)
[51] $[52]$	QIAN, J. Closure approach to high-order structure functions of turbulence. <i>Physical Review L</i> , 36 (6), 7525 (1996) QIAN, J. Closure approach to high-order structure functions of turbulence. <i>Physical Review Letters</i> , 84 (4), 646–654 (2000)
[53]	MCCOMB, W. D. Homogeneous, Isotropic Turbulence: Phenomenology, Renormalization and Statistical Closures, Oxford University Press, Oxford (2014)
[54]	SHE, Z. S., CHEN, X., and HUSSAIN, F. Quantifying wall turbulence via a symmetry approach: a lie group theory. <i>Journal of Fluid Mechanics</i> , 827 , 322–356 (2017)
[55]	CHEN, X., HUSSAIN, F., and SHE, Z. S. Quantifying wall turbulence via a symmetry approach, part 2, Reynolds stresses. <i>Journal of Fluid Mechanics</i> , 850 , 401–438 (2018)
[56]	CHEN, X. and SHE, Z. S. Analytic prediction for planar turbulent boundary layers. <i>Science China Physics, Mechanics and Astronomy</i> , 59 , 114711 (2016)
[57]	SHE, Z. S., WU, Y., CHEN, X., and HUSSAIN, F. A multi-state description of roughness effects in turbulent pipe flow. <i>New Journal of Physics</i> , 14 (9), 093054 (2012)
[58]	ZHANG, Y. S., BI, W. T., HUSSAIN, F., LI, X. L., and SHE, Z. S. Mach-number-invariant mean-velocity profile of compressible turbulent boundary layers. <i>Physical Review Letters</i> , 109 (5), 054502 (2012)
[59]	WU, B., BI, W. T., HUSSAIN, F., and SHE, Z. S. On the invariant mean velocity profile for compressible turbulent boundary layers. <i>Journal of Turbulence</i> , 18 (182), 186–202 (2017)
[60]	LUO, J. S., WANG, X. J., and ZHOU, H. Inherent mechanism of breakdown in laminar-turbulent transition of plane channel flows. <i>Science in China Series G: Physics Mechanics and Astronomy</i> , 48 (2), 228–236 (2005)
[61]	TANG, H. T., LUO, J. S., and ZHOU, H. Mechanism of breakdown in laminar-turbulent transi- tion of incompressible boundary layer on a flat plate. <i>Transactions of Tianjin University</i> , 13 (2), 79–87 (2007)
[62]	HUANG, Z. F., WEI, C., and ZHOU, H. The mechanism of breakdown in laminar-turbulent transition of a supersonic boundary layer on a flat plate-temporal mode. <i>Science in China</i> , 48 (5), 614–625 (2005)
[63]	CAO, W., HUANG, Z. F., and ZHOU, H. Study of the mechanism of breakdown in laminar- turbulent transition of a supersonic boundary layer on a flat plate. <i>Applied Mathematics and</i> <i>Mechanics (English Edition)</i> , 27 (4), 425–434 (2006) https://doi.org/10.1007/s10483-006-0401-1
[64]	DONG, M. and ZHOU, H. A simulation on bypass transition and its key mechanism. <i>Science China (Physics, Mechanics and Astronomy)</i> , 56 (4), 775–784 (2013)
[65]	ZHANG, Y. M. and ZHOU, H. Numerical study of local boundary layer receptivity to freestream vortical disturbances. <i>Applied Mathematics and Mechanics (English Edition)</i> , 26 (5), 547–554 (2005) https://doi.org/10.1007/BF02466327
[66]	GAO, J., LUO, J. S., and WU, X. S. Receptivity of hypersonic boundary layer due to fast-slow acoustics interaction. <i>Acta Mechanica Sinica</i> , 31 (6), 899–909 (2015)
[67]	REN, J. and FU, S. Study of the discrete spectrum in a Mach 4.5 Görtler flow. <i>Flow Turbulence</i> and Combustion, 94 (2), 339–357 (2015)
[68]	JIANG, X. Y. and LEE, C. B. Review of research on the receptivity of hypersonic boundary layer. <i>Journal of Experiments in Fluid Mechanics</i> , 31 (2), 1–11 (2017)
[69]	REN, J. and FU, S. Secondary instabilities of Görtler vortices in high-speed boundary layer flows. <i>Journal of Fluid Mechanics</i> , 781 , 388–421 (2015)
[70]	REN, J., FU, S., and HANIFI, A. Stabilization of the hypersonic boundary layer by finite-amplitude streaks. <i>Physics of Fluids</i> , 28 (2), 754–203 (2016)
[71]	SU, C. H. and ZHOU, H. Transition prediction for supersonic and hypersonic boundary layers on a cone with angle of attack. <i>Science in China Series G: Physics, Mechanics and Astronomy</i> , 52 (8), 1223–1232 (2009)

Bohua SUN

208

- [72] SU, C. H. and ZHOU, H. Transition prediction of the supersonic boundary layer on a cone under the consideration of receptivity to slow acoustic waves. *Science China Physics, Mechanics and Astronomy*, 54(10), 1875–1882 (2011)
- [73] SU, C. H. The reliability of the improved e^N method for the transition prediction of boundary layers on a flat plate. *Science China Physics, Mechanics and Astronomy*, **55**(5), 837–843 (2012)
- [74] SU, C. H. Physical implication of two problems in transition prediction of boundary layers based on linear stability theory. *Science China Physics, Mechanics and Astronomy*, 57(5), 950–962 (2014)
- [75] LI, X., FU, D. X., and MA, Y. W. Direct numerical simulation of hypersonic boundary layer transition over a blunt cone. AIAA Journal, 46(11), 2899–2913 (2008)
- [76] LIANG, X., LI, X. L., FU, D. X., and MA, Y. W. Effects of wall temperature on boundary layer stability over a blunt cone at Mach 7.99. *Computers and Fluids*, **39**(2), 359–371 (2010)
- [77] SU, C. H. and ZHOU, H. The variation of transition location in response to the variation of the amplitudes of initial disturbances. *Science China Physics, Mechanics and Astronomy*, 53(6), 1109–1115 (2010)
- [78] LIU, J. X. and LUO, J. S. Effect of disturbances at inlet on hypersonic boundary layer transition on a blunt cone at small angle of attack. *Applied Mathematics and Mechanics (English Edition)*, **31**(5), 535–544 (2010) https://doi.org/10.1007/s10483-010-0501-z
- [79] LI, J. and LUO, J. S. Applications of parabolized stability equation for predicting transition position in boundary layers. Applied Mathematics and Mechanics (English Edition), 33(6), 679– 686 (2012) https://doi.org/10.1007/s10483-012-1579-7
- [80] ZHANG, Y. M. and ZHOU, H. PSE as applied to problems of transition in compressible boundary layers. Applied Mathematics and Mechanics (English Edition), 29(7), 833–840 (2008) https://doi.org/10.1007/s10483-008-0701-8
- [81] ZHANG, Y. M. and ZHOU, H. PSE as applied to problems of secondary instability in supersonic boundary layers. Applied Mathematics and Mechanics (English Edition), 29(1), 1–8 (2008) https://doi.org/10.1007/s10483-008-0101-7
- [82] ZHANG, Y. M. and SU, C. H. Self-consistent parabolized stability equation (PSE) method for compressible boundary layer. Applied Mathematics and Mechanics (English Edition), 36(7), 835– 846 (2015) https://doi.org/10.1007/s10483-015-1951-9
- [83] YU, G. T., GAO, J., and LUO, J. S. Stability analysis method considering non-parallelism: EPSE method and its application. Applied Mathematics and Mechanics (English Edition), 37(1), 27–36 (2016) https://doi.org/10.1007/s10483-016-2013-9
- [84] GUO, L. L., TANG, D. B., and LIU, J. X. Evolution analysis of TS wave and high-order harmonic waves in boundary layers. *Transactions of Nanjing University of Aeronautics & Astronautics*, 23, 8–14 (2006)
- [85] TANG, D. B., MA, Q. R., and CHENG, G. W. Study of compressible nonparallel flow stability for boundary layers. Acta Aeronautica et Astronautica Sinica, 23(2), 166–169 (2002)
- [86] XIA, H., TANG, D. B., and LU, C. G. Nonparallel stability analysis of three dimensional disturbance wave in boundary layers. Acta Mechanica Sinica, 34(5), 688–695 (2002)
- [87] GUO, X., TANG, D. B., and SHEN, Q. Boundary layer stability with multiple modes in hypersonic flows. *Modern Physics Letters B*, 23, 321–324 (2009)
- [88] LIU, J. X., TANG, D. B., and YANG, Y. Z. On nonlinear-evolution of C-type instability in nonparallel boundary layers. *Chinese Journal of Aeronautics*, **20**(4), 313–319 (2007)
- [89] TANG, D. B. Boundary Layer Transition, Science Press, Beijing (2015)
- [90] LEE, C. B. and CHEN, S. Y. A review of recent progress in the study of transition in hypersonic boundary layer. *National Science Review*, nwy052 (2018)
- [91] ZHANG, C. H., TANG, Q., and LEE, C. B. Hypersonic boundary-layer transition on a flared cone. Acta Mechanica Sinica, 29(1), 48–54 (2013)
- [92] ZHANG, C. H. and LEE, C. B. Rayleigh-scattering visualization of the development of secondmode waves. *Journal of Visualization*, 20(1), 1–6 (2016)

[93]	ZHANG, C. H., ZHU, Y. D., CHEN, X., YUAN, H. J., WU, J. Z., CHEN, S. Y., LEE, C. B.,
	and GAD-EL-HAK, M. Transition in hypersonic boundary layers. AIP Advances, 5(10), 107137
	(2015)

- [94] ZHU, Y. D., ZHANG, C. H., CHEN, X., YUAN, H. J., WU, J. Z., CHEN, S. Y., LEE, C. B., and GAD-EL-HAK, M. Transition in hypersonic boundary layers: role of dilatational waves. AIAA Journal, 54, 3039–3049 (2016)
- [95] ZHU, Y. D., CHEN, X., WU, J. Z., CHEN, S. Y., LEE, C. B., and GAD-EL-HAK, M. Aerodynamic heating in transitional hypersonic boundary layers: role of second-mode instability. *Physics* of Fluids, **30**(1), 011701 (2018)
- [96] ZHU, Y. D., LEE, C. B., and CHEN, X. Newly identified principle for aerodynamic heating in hypersonic flows. *Journal of Fluid Mechanics*, 855, 152–180 (2018)
- [97] SUN, B. H. and ORAN, E. S. New principle for aerodynamic heating. *National Science Review*, 5, 606–607 (2018)
- [98] CHEN, X., ZHU, Y. D., and LEE, C. B. Interactions between second mode and low-frequency waves in a hypersonic boundary layer. *Journal of Fluid Mechanics*, 820, 693–735 (2017)
- [99] LI, X. L., MA, Y. W., and FU, D. X. High efficient method for incompressible N-S equations and analysis of two-dimensional turbulent channel flow (in Chinese). Acta Mechanica Sinica, 33(5), 577–587 (2001)
- [100] LI, X. L., MA, Y. W., and FU, D. X. Scaling law in two-dimensional turbulent channel flow (in Chinese). Acta Mechanica Sinica, 34(4), 604–608 (2002)
- [101] LI, X. L., FU, D. X., and MA, Y. W. Direct numerical simulation of a spatially evolving supersonic turbulent boundary layer at Ma = 6. Chinese Physics Letters, 23(6), 1519–1522 (2006)
- [102] LI, X. L., FU, D. X., and MA, Y. W. Assessment of the compressible turbulence model by using the DNS data (in Chinese). *Chinese Journal of Theoretical and Applied Mechanics*, 44, 222–229 (2012)
- [103] DENG, X. G. and ZHANG, H. X. Developing high-order weighted compact nonlinear schemes. Journal of Computational Physics, 165(1), 22–44 (2000)
- [104] MA, Y. W. and FU, D. X. Fourth order accurate compact scheme with group velocity control (GVC). Science in China Series A: Mathematics, 44(9), 1197–1204 (2001)
- [105] DONG, M., LUO, J. S., and CAO, W. Numerical investigation of the evolution of disturbances in supersonic sharp cone boundary layers. *Applied Mathematics and Mechanics (English Edition)*, 27(6), 713–719 (2006) https://doi.org/10.1007/s10483-006-0601-1
- [106] DONG, M. and LUO, J. S. Mechanism of transition in a hypersonic sharp cone boundary layer with zero angle of attack. Applied Mathematics and Mechanics (English Edition), 28(8), 1019– 1028 (2007) https://doi.org/10.1007/s10483-007-0804-2
- [107] DONG, M. and ZHOU, H. Inflow boundary condition for DNS of turbulent boundary layers on supersonic blunt cones. Applied Mathematics and Mechanics (English Edition), 29(8), 985–998 (2008) https://doi.org/10.1007/s10483-008-0802-3
- [108] DONG, M. and LUO, J. S. The influence of the turbulent statistical characteristics by cone effect in a supersonic boundary layer (in Chinese). *Chinese Journal of Theoretical and Applied Mechanics*, 40(3), 394–401 (2008)
- [109] DONG, M. Direct numerical simulation of a spatially evolving hypersonic blunt cone turbulent boundary layer at Mach 6 (in Chinese). Acta Aerodynamica Sinica, 27(2), 199–205 (2009)
- [110] DONG, M. and ZHOU, H. Improvement of turbulent mode in the calculation of aerodynamic heat in hypersonic turbulent boundary layer (in Chinese). Scientia Sinica Physica, Mechanica and Astronomica, 40(2), 231–241 (2010)
- [111] LI, Q. B. and FU, S. High-order accurate gas-kinetic scheme and turbulence simulation (in Chinese). Scientia Sinica Physica, Mechanica and Astronomica, 44(3), 278–284 (2014)
- [112] LI, Q. B., TAN, S., FU, S., and XU, K. Numerical simulation of compressible turbulence with gas-kinetic BGK scheme. 13th Asian Congress of Fluid Mechanics, Dhaka, Bangladesh, 12–17 (2010)

- [113] TAN, S., LI, Q. B., FU, S., and ZENG, S. Engineering simulation of turbulence with gas-kinetic BGK scheme. AIP Conference Proceedings, 1376, 78–80 (2011)
- [114] FU, S. and LI, Q. B. Numerical simulation of compressible mixing layers. International Journal of Heat and Fluid Flow, 27(5), 895–901 (2006)
- [115] FU, S. and GUO, Y. Study of wall-parameter-free low-Reynolds-number nonlinear eddy-viscosity model (in Chinese). Acta Mechanica Sinica, 33(2), 145–152 (2001)
- [116] ZHANG, G. H. and FU, S. Second-moment closure modelling of the compressible homogeneous turbulent shear flow (in Chinese). Acta Mechanica Sinica, 32(2), 141–150 (2000)
- [117] KRAICHNAN, R. H. and CHEN, S. Y. Is there a statistical mechanics of turbulence? *Physica D: Nonlinear Phenomena*, **37**(1-3), 160–172 (1989)
- [118] MENEVEAU, C. Statistics of turbulence subgrid-scale stresses: necessary conditions and experimental tests. *Physics of Fluids*, 6(2), 815–833 (1994)
- [119] SHI, Y. P., XIAO, Z. L., and CHEN, S. Y. Constrained subgrid-scale stress model for large eddy simulation. *Physics of Fluids*, **20**(1), 011701 (2008)
- [120] MENEVEAU, C. Germano identity-based subgrid-scale modeling: a brief survey of variations on a fertile theme. *Physics of Fluids*, 24(12), 121301 (2012)
- [121] CHEN, S. Y., XIA, Z. H., PEI, S. Y., WANG, J. C., YANG, Y. T., XIAO, Z. L., and SHI, Y. P. Reynolds-stress-constrained large-eddy simulation of wall-bounded turbulent flows. *Journal of Fluid Mechanics*, **703**, 1–28 (2012)
- [122] CHEN, S. Y., CHEN, Y. C., XIA, Z. H., QU, K., SHI, Y. P., XIAO, Z. L., LIU, Q. H., CAI, Q. D., LIU, F., LEE, C. B., ZHANG, R. K., and CAI, J. S. Constrained large-eddy simulation and detached eddy simulation of flow past a commercial aircraft at 14 degrees angle of attack. *Science China Physics, Mechanics and Astronomy*, 56(2), 270–276 (2013)
- [123] CHEN, S. Y., WANG, M. R., and XIA, Z. H. Multiscale fluid mechanics and modeling. Proceedia IUTAM, 10, 100–114 (2014)
- [124] LI, J. C. and XIE, Z. T. Large-eddy simulation for canopy turbulent flow (in Chinese). Acta Mechanica Sinica, 31(4), 406–415 (1999)
- [125] LI, J. C. Large eddy simulation of complex turbulent flows: physical aspects and research trends. Acta Mechanica Sinica, 17(4), 289–301 (2001)
- [126] CUI, G. X., SHI, R. F., WANG, Z. S., XU, C. X., and ZHANG, Z. S. Large eddy simulation of city micro-atmospheric environment. *Science in China Series G: Physics, Mechanics and Astronomy*, 51(8), 933–944 (2008)
- [127] CUI, G. X., XU, C. X., FANG, L., SHAO, L., and ZHANG, Z. S. A new subgrid eddy-viscosity model for large-eddy simulation of anisotropic turbulence. *Journal of Fluid Mechanics*, 582, 377–397 (2007)
- [128] LU, X. Y. and DALTON, C. Calculation of the timing of vortex formation from an oscillating cylinder. Journal of Fluids and Structures, 10(5), 527–541 (1996)
- [129] LU, X. Y. and SATO, J. A numerical study of flow past a rotationally oscillating circular cylinder. Journal of Fluids and Structures, 10(8), 829–849 (1996)
- [130] LU, X. Y., DALTON, C., and ZHANG, J. F. Application of large eddy simulation to an oscillating flow past a circular cylinder. *Journal of Fluids Engineering*, **119**(3), 519–525 (1997)
- [131] ZHOU, Y. and ANTONIA, R. A. A study of turbulent vortices in the near wake of a cylinder. Journal of Fluid Mechanics, 253, 643–661 (1993)
- [132] ZHOU, Y. and ANTONIA, R. A. Critical points in a turbulent near wake. Journal of Fluid Mechanics, 275, 59–81 (1994)
- [133] ZHANG, H. J., ZHOU, Y., and ANTONIA, R. A. Longitudinal and spanwise vortical structures in a turbulent near wake. *Physics of Fluids*, **12**(11), 2954–2964 (2000)
- [134] RINOSHIKA, A. and ZHOU, Y. Orthogonal wavelet multi-resolution analysis of a turbulent cylinder wake. *Journal of Fluid Mechanics*, **524**, 229–248 (2005)
- [135] WANG, H. F., ZHOU, Y., CHAN, C. K., and LAM, K. S. Effect of initial conditions on interaction between a boundary layer and a wall-mounted finite-length-cylinder wake. *Physics of Fluids*, 18(6), 065106 (2006)

[136]	XU, S. J., ZHOU, Y., and WANG, M. H. A symmetric binary-vortex street behind a longitudi- nally oscillating cylinder. <i>Journal of Fluid Mechanics</i> , 556 , 27–43 (2006)
[137]	ALAM, M. D. M. and ZHOU, Y. Turbulent wake of an inclined cylinder with water running. <i>Journal of Fluid Mechanics</i> , 589 , 261–303 (2007)
[138]	HU, J. C. and ZHOU, Y. Flow structure behind two staggered circular cylinders, part 2, heat and momentum transport. <i>Journal of Fluid Mechanics</i> , 607 , 81–107 (2008)
[139]	HU, J. C. and ZHOU, Y. Flow structure behind two staggered circular cylinders, part 1, down-stream evolution and classification. <i>Journal of Fluid Mechanics</i> , 607 , 51–80 (2008)
[140]	ZHOU, T., WANG, H., RAZALI, S. M., ZHOU, Y., and CHENG, L. Three-dimensional vorticity measurements in the wake of a yawed circular cylinder. <i>Physics of Fluids</i> , $22(1)$, 015108 (2010)
[141]	BAI, Y., JIA, Y. X., LEE, C. B., and ZHU, Y. D. Experimental study of a periodical flapping flag. Acta Physica Sinica, 65 (12), 178–184 (2016)
[142]	ZHANG, M. M., CHENG, L., and ZHOU, Y. Closed-loop-controlled vortex shedding and vibra- tion of a flexibly supported square cylinder under different schemes. <i>Physics of Fluids</i> , 16 (5), 1439–1448 (2004)
[143]	WANG, H. F. and ZHOU, Y. The finite-length square cylinder near wake. <i>Journal of Fluid Mechanics</i> , 638 , 453–490 (2009)
[144]	ALAM, M. M., ZHOU, Y., and WANG, X. W. The wake of two side-by-side square cylinders. <i>Journal of Fluid Mechanics</i> , 669 , 432–471 (2011)
[145]	XIA, Z. H., SHI, Y. P., CAI, Q. D., and GAI, J. Dissipation function in turbulent plane Poiseuille and Couette flows subject to spanwise rotations. <i>Applied Mathematics and Mechanics (English Edition)</i> , 40 (2), 185–192 (2019) https://doi.org/10.1007/s10483-019-2422-6
[146]	DONG, G. J. and LU, X. Y. Characteristics of flow over traveling wavy foils in a side-by-side arrangement. <i>Physics of Fluids</i> , $19(5)$, 057107 (2007)
[147]	LU, X. Y. and LIAO, Q. Dynamic responses of a two-dimensional flapping foil motion. <i>Physics of Fluids</i> , 18 (9), 098104 (2006)
[148]	LU, X. Y. and YIN, X. Z. Propulsive performance of a fish-like travelling wavy wall. Acta Mechanica, $175(1-4)$, $197-215$ (2005)
[149]	LEE, C. B. and WU, J. Z. Transition in wall-bounded flows. <i>Applied Mechanics Reviews</i> , 61 (3), 030802 (2008)
[150]	LEE, C. B. New features of CS solitons and the formation of vortices. <i>Physics Letters A</i> , $247(6)$, 397–402 (1998)
[151]	LEE, C. B. and WANG, S. Study of the shock motion in a hypersonic shock system/turbulent boundary layer interaction. <i>Experiments in Fluids</i> , $19(3)$, 143–149 (1995)
[152]	LEE, C. B., HONG, Z. X., KACHANOV, Y. S., BORODULIN, V., and GAPONENKO, V. A study in transitional flat plate boundary layers: measurement and visualization. <i>Experiments in Fluids</i> , 28 (3), 243–251 (2000)
[153]	LEE, C. B. Possible universal transitional scenario in a at plate boundary layer: measurement and visualization. <i>Physical Review E</i> , $62(3)$, 3659–3670 (2000)
[154]	HAMA, F. R. and NUTANT, J. Detailed flow-field observations in the transition process in a thick boundary layer. <i>Proceedings of 1963 Heat Transfer and Fluid Mechanics</i> , Stanford University Press, Stanford (1963)
[155]	LEE, C. B. CS-soliton: its feature and importance (in Chinese). Experiments and Measurements in Fluid Mechanics, 17(3), 57–60 (2003)
[156]	LEE, C. B. and LI, R. Q. Dominant structure for turbulent production in a transitional boundary layer. <i>Journal of Turbulence</i> , 8 , N55 (2007)
[157]	LEE, C. B. On the formation of the streamwise vortex in a transitional boundary layer. Acta Physica Sinica, $50(1)$, 182–184 (2001)
[158]	ZHAO, Y. M., YANG, Y., and CHEN, S. Y. Evolution of material surfaces in the temporal transition in channel flow. <i>Journal of Fluid Mechanics</i> , 793 , 840–876 (2016)

Bohua SUN

212

- [159] LIAN, Q. X. A visual study of the coherent structure of the turbulent boundary layer in flow with adverse pressure gradient. *Journal of Fluid Mechanics*, **215**, 101–124 (1990)
- [160] LIAN, Q. X. A kind of fast changing coherent structure in a turbulent boundary layer. Acta Mechanica Sinica, 15(3), 193–200 (1999)
- [161] GUO, H., LIAN, Q. X., LI, Y., and WANG, H. W. A visual study on complex flow structures and flow breakdown in a boundary layer transition. *Experiments in Fluids*, **37**(3), 311–322 (2004)
- [162] GUO, H., LI, X. H., WANG, H. W., and FENG, Y. L. Experimental study on boundary-layer transition control by spanwise discrete suction. *Journal of Experiments in Fluid Mechanics*, 28(6), 13–19 (2014)
- [163] GONG, A. L., LI, R. Q., and LEE, C. B. On the generation of low-frequency signals in a transitional boundary layer. Acta Physica Sinica, 51(5), 1068–1074 (2002)
- [164] MENG, Q. G., LI, R. Q., and LEE, C. B. A link between the turbulent cascade and the dynamics of transition. Acta Physica Sinica, 53(8), 2621–2624 (2004)
- [165] MENG, Q. G., CAI, Q. D., and LEE, C. B. Limitation for the energy spectrum analysis for the understanding of the turbulent cascade. Acta Physica Sinica, 53(9), 3090–3093 (2004)
- [166] LI, R. Q. and LEE, C. B. A link between chaos dynamics and the onset of turbulence in a transitional boundary layer. Acta Physica Sinica, 51(8), 1743–1749 (2002)
- [167] LI, R. Q. and LEE, C. B. Problems in the paper "a link between chaos dynamics and the onset of turbulence in a transitional boundary layer". Acta Physica Sinica, 54(1), 1743–1749 (2005)
- [168] GAO, Q., ORTIZ-DUENAS, C., and LONGMIRE, E. Circulation signature of vortical structures in turbulent boundary layers. 16th Australasian Fluid Mechanics Conference (AFMC), The University of Queensland, Brisbane, 135–141 (2007)
- [169] GAO, Q., ORTIZ-DUENAS, C., and LONGMIRE, E. Analysis of vortex populations in turbulent wall-bounded flows. *Journal of Fluid Mechanics*, 678, 87–123 (2011)
- [170] GAO, Q., ORTIZ-DUENAS, C., and LONGMIRE, E. Evolution of coherent structures in turbulent boundary layers based on moving tomographic PIV. *Experiments in Fluids*, 54(12), 1625 (2013)
- [171] WANG, J. J., ZHANG, C., and PAN, C. Effects of roughness elements on bypass transition induced by a circular cylinder wake. *Journal of Visualization*, 14(1), 53–61 (2011)
- [172] ZHANG, C., PAN, C., and WANG, J. J. Evolution of vortex structure in boundary layer transition induced by roughness elements. *Experiments in Fluids*, 51(5), 1343–1352 (2011)
- [173] HE, G. S., WANG, J. J., and PAN, C. Initial growth of a disturbance in a boundary layer influenced by a circular cylinder wake. *Journal of Fluid Mechanics*, **718**, 116–130 (2013)
- [174] PAN, C., WANG, J. J., and HE, G. S. Experimental investigation of wake-induced bypass transition control by surface roughness. *Chinese Physics Letters*, 29(10), 104704 (2012)
- [175] PAN, C., WANG, J. J., and ZHANG, C. Identification of Lagrangian coherent structures in the turbulent boundary layer. Science in China Series G: Physics, Mechanics and Astronomy, 52(2), 248–257 (2009)
- [176] PAN, C., WANG, J. J., ZHANG, P. F., and FENG, L. H. Coherent structures in bypass transition induced by a cylinder wake. *Journal of Fluid Mechanics*, 603, 367–389 (2008)
- [177] HE, G. S., PAN, C., FENG, L. H., GAO, Q., and WANG, J. J. Evolution of Lagrangian coherent structures in a cylinder-wake disturbed at plate boundary layer. *Journal of Fluid Mechanics*, 792, 274–306 (2016)
- [178] FENG, L. H., WANG, J. J., and PAN, C. Proper orthogonal decomposition analysis of vortex dynamics of a circular cylinder under synthetic jet control. *Physics of Fluids*, 23(1), 014106 (2011)
- [179] WANG, J. J., PAN, C., and ZHANG, P. F. On the instability and reproduction mechanism of a laminar streak. *Journal of Turbulence*, 10, N26 (2009)
- [180] ZHANG, Z., SUN, Z. L., and YAN, D. C. Experimental research on the mechanism of reverse transition in turbulent boundary layer. Acta Scientiarum Naturalium-Universitatis Pekinensis, 42(3), 305–309 (2006)

214	Bohua SUN
[181]	FAN, Q. L., WANG, L. T., ZHANG, H. Q., GUO, Y. C., and LIN, W. Y. Coherent structures in turbulent round jets. <i>Advances in Mechanics</i> , 32 , 109–118 (2002)
[182]	FAN, Q. L., ZHANG, H. Q., GUO, Y. C., and WANG, X. L. Particle-vortex interactions in turbulent shear flows. <i>Advances in Mechanics</i> , 31 , 611–620 (2001)
[183]	WANG, J. J. and XIA, K. Q. Advances in experimental investigation of Rayleigh-Benard turbu- lent convection. <i>Advances in Mechanics</i> , 29 , 557–566 (1999)
[184]	WANG, S. F., JIANG, F., NIU, Z. N., and WU, Z. Z. An experimental study on turbulent coherent structures near a sheared air-water interface. <i>Acta Mechanica Sinica</i> , 15 (4), 289–298 (1999)
[185]	ZHONG, H. J., CHEN, S. Y., and LEE, C. B. Experimental study of freely falling thin disks: transition from planar zigzag to spiral. <i>Physics of Fluids</i> , 23 (1), 011702 (2011)
[186]	LEE, C. B., SU, Z., ZHONG, H., CHEN, S., ZHOU, M., and WU, J. Experimental investigation of freely falling thin disks, part 2, transition of three-dimensional motion from zigzag to spiral. <i>Journal of Fluid Mechanics</i> , 732 , 77–104 (2013)
[187]	JIA, L. C., ZHU, Y. D., JIA, Y. X., YUAN, H. J., and LEE, C. B. Image pre-processing method for near-wall PIV measurements over moving curved interfaces. <i>Measurement Science and Technology</i> , 28 (3), 035201 (2017)
[188]	JIA, L. C., ZOU, T. D., ZHU, Y. D., and LEE, C. B. Rotor boundary layer development with inlet guide vane (IGV) wake impingement. <i>Physics of Fluids</i> , 30 (4), 040911 (2018)
[189]	LI, C. and ZHANG, Y. C. Effect of glow discharge on hypersonic flat plate boundary layer. <i>Applied Mathematics and Mechanics (English Edition)</i> , 40 (2), 249–260 (2019) https://doi.org/10.1007/s10483-019-2424-9
[190]	ZHANG, Y. C. and LI, C. Transition control of Mach 6.5 hypersonic flat plate boundary layer. <i>Applied Mathematics and Mechanics (English Edition)</i> , 40 (2), 283–292 (2019) https://doi.org/ 10.1007/s10483-019-2423-8
[191]	ZHU, Y. D., YUAN, H. J., ZHANG, C. H., and LEE, C. B. Image-preprocessing method for near- wall particle image velocimetry (PIV) image interrogation with very large in-plane displacement. <i>Measurement Science and Technology</i> , 24 (12), 125302 (2013)
[192]	ZHU, Y. D. Experimental and numerical study of flow structures of the second-mode instability. <i>Applied Mathematics and Mechanics (English Edition)</i> , 40 (2), 273–282 (2019) https://doi.org/ 10.1007/s10483-019-2430-9
[193]	YI, S. H., TIAN, L. F., ZHAO, Y. X., and HE, L. The new advance of the experimental research on compressible turbulence based on the NPLS technique. <i>Advances in Mechanics</i> , 41 , 379–390 (2011)
[194]	HUANG, N. and ZHENG, X. J. A laboratory test of the electrification phenomenon in windblown sand flux. <i>Chinese Science Bulletin</i> , 46 (5), 417–420 (2001)
[195]	ZHENG, X. J., HUANG, N., and ZHOU, Y. H. Laboratory measurement of electrification of wind-blown sands and simulation of its effect on sand saltation movement. <i>Journal of Geophysical Research: Atmospheres</i> , 108 (D10), 4322 (2003)
[196]	ZHENG, X. J., ZHANG, R., and HUANG, H. J. Theoretical modeling of relative humidity on contact electrification of sand particles. <i>Scientific Reports</i> , 4 , 4399 (2014)
[197]	ZHENG, X. J. and WANG, P. Numerical simulation on stochastic movement of sands in wind- blown sand. <i>Journal of Desert Research</i> , 26 (2), 184–188 (2006)
[198]	BO, T. L. and ZHENG, X. J. Numerical simulation of the evolution and propagation of aeolian dune fields toward a desert-oasis zone. <i>Geomorphology</i> , 180 , 24–32 (2013)
[199]	CHEN, X. R., DENG, X. Y., WANG, Y. K., LIU, P. Q., and GU, Z. F. Influence of nose perturbations on behaviors of asymmetric vortices over slender body. <i>Acta Mechanica Sinica</i> , 18 (6), 581–593 (2002)